Author’s response to reviews

Title: Mangosteen (Garcinia mangostana) flesh supplementation attenuates biochemical and morphological changes in the liver and kidney of high fat diet-induced obese rats

Authors:
Noratirah Shazlin Muhamad Adyab (atirshaz_wanmie@yahoo.com)
Asmah Rahmat (profasmah@gmail.com)
Noor Atiqah Aizan Abdul Kadir (atiqahaizan@yahoo.com)
Hawa Jaafar (hawazej@upm.edu.my)
Radhiah Shukri (radhiah@upm.edu.my)
Nurul Shazini Ramli (shazini@upm.edu.my)

Version: 1 Date: 15 Feb 2018

Author’s response to reviews:

Dear Editor,

I am resubmitting a manuscript entitled “Protective effects of mangosteen (Garcinia mangostana) flesh on biochemical and morphological changes in the liver and kidney of high fat diet-induced obese rats” for publication in BMC Complementary & Alternative Medicine. The corrections made according to the comments by the Editor.

The authors' response letter has been included as a supplementary file and are attached under the item named “Response to Editorial Comments”.

The authors declare no conflict of interest and this manuscript has not been submitted elsewhere for publication and there are none related manuscript pertaining this study under review, in process or have been published. I declare all the co-authors have agreed for submission to BMC Complementary & Alternative Medicine and that the MS is prepared strictly according to the Journal format as provided in the instruction to authors. The authors have declared that no competing interests exist.

On behalf of the co-authors, I would like to thank you in advance for the time and effort you expend considering this manuscript for publication.

Yours sincerely,
Nurul Shazini Ramli
Universiti Putra Malaysia, Malaysia.